Author's response to reviews

Title: Association between TGFBR1*6A and osteosarcoma: A Chinese case-control study

Authors:

Yunsheng Hu (huys111@hotmail.com)
Yong Pan (panyong20@yahoo.com)
Wenhai Li (liwenhai@fmmu.edu.cn)
Yong Zhang (tdgk09@fmmu.edu.cn)
Jun Li (lijun67@fmmu.edu.cn)
Baoan Ma (huys11@sina.com)

Version: 3 Date: 7 April 2010

Author's response to reviews: see over
Dear Sabina:

We agree to that the manuscript needs language correction before being published. Thank you for your advice and providing the website of the Manuscript Presentation Service. The manuscript has been revised by the Manuscript Presentation Service (www.biomedes.co.uk). All changes made are highlighted with tracked changes.

One reference (reference 15) we cited was not correct, and we replaced it with the right reference. The reference is blue coloured.

We hope the revised manuscript can be accepted by BMC cancer.

With best wishes,

Bao-An Ma, MD, PhD
Director of Orthopaedic Oncology Institute of PLA, Tangdu Hospital, Fourth Military Medical University, Xi’an, 710038, China.